

Modelling cyber-security experts' decision making processes using aggregation operators



Simon Miller, Christian Wagner, Uwe Aickelin, Jonathan M. Garibaldi *

Intelligent Modelling and Analysis (IMA) Research Group, Lab for Uncertainty in Data and Decision Making (LUCID), Horizon Digital Economy Research, School of Computer Science, University of Nottingham, Nottingham NG8 1BB, UK

ARTICLE INFO

Article history: Received 14 December 2015 Received in revised form 20 June 2016 Accepted 3 August 2016 Available online 10 August 2016

Keywords: Expert decision making Cyber-security Evolutionary algorithm Ordered weighted average Weighted average

ABSTRACT

An important role carried out by cyber-security experts is the assessment of proposed computer systems, during their design stage. This task is fraught with difficulties and uncertainty, making the knowledge provided by human experts essential for successful assessment. Today, the increasing number of progressively complex systems has led to an urgent need to produce tools that support the expert-led process of system-security assessment. In this research, we use weighted averages (WAs) and ordered weighted averages (OWAs) with evolutionary algorithms (EAs) to create aggregation operators that model parts of the assessment process. We show how individual overall ratings for security components can be produced from ratings of their characteristics, and how these individual overall ratings can be aggregated to produce overall rankings of potential attacks on a system. As well as the identification of salient attacks and weak points in a prospective system, the proposed method also highlights which factors and security components contribute most to a component's difficulty and attack ranking respectively. A real world scenario is used in which experts were asked to rank a set of technical attacks, and to answer a series of questions about the security components that are the subject of the attacks. The work shows how finding good aggregation operators, and identifying important components and factors of a cyber-security problem can be automated. The resulting operators have the potential for use as decision aids for systems designers and cyber-security experts, increasing the amount of assessment that can be achieved with the limited resources available.

© 2016 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

1. Introduction

As Internet use becomes ever more pervasive in day-to-day life for tasks including Internet banking, e-commerce and e-government, the risk of cyber-crime is a growing concern (see Detica and Office of Cyber Security and Information Assurance, 2011, Anderson et al., 2013 and Cruz et al., 2014). Assessing security risks associated with proposed systems in their design phase is a non-trivial task that involves managing multiple sources of uncertain information. For example, it is very difficult to estimate the costs of a successful attack, the likelihood of a rare attack, as the tools and technologies available to attackers/defenders are constantly changing (Tregear, 2001). Because of these difficulties, a great deal of expertise is required to carry out such assessments. Typically, cyber-security

* Corresponding author.

E-mail addresses: s.miller@nottingham.ac.uk (S. Miller), christian.wagner@nottingham.ac.uk (C. Wagner), uwe.aickelin@nottingham.ac.uk (U. Aickelin), jon.garibaldi@nottingham.ac.uk (J.M. Garibaldi).

http://dx.doi.org/10.1016/j.cose.2016.08.001

^{0167-4048/© 2016} The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http:// creativecommons.org/licenses/by/4.0/).

experts are employed as they can provide comprehensive assessments based on considerable expertise and insight while also being able to assess the viability of existing cybersecurity tools and processes (e.g., anti-malware software). From a computational point of view, their work can be likened to that of a highly complex aggregation function, considering a large number of uncertain data sources (e.g., other experts, users, systems designers and security software/hardware) and fusing these sources to build overall security assessments. However, there is often a shortage of the level of cyber-security expertise required to carry out detailed assessments, leading to an urgent requirement for techniques and tools that can be used to support experts, reducing their workload, and systems designers, using models of expert knowledge to obtain estimates of security for system designs.

As stated previously, the job of performing security assessments is comparable to a complex aggregation function, fusing multiple sources of disparate data to form an overall assessment. In order to replicate this process in a computational model, aggregation operators such as the arithmetic mean, weighted average (WA), and ordered weighted average (OWA) could be considered. As will be shown later, WAs and OWAs allow the application of weightings to specific objects and specific positions in an ordering, respectively. We will employ both of these methods in this research, exploiting their characteristics to produce fused assessments. WAs are used to compute assessments of security components using sub-assessments of their characteristics, and OWA operators are used to compute salience/difficulty rankings of specific technical attacks using security component assessments. A difficulty when using WAs and OWAs is finding suitable weightings for a particular task, as there are an near-infinite number of possibilities. Evolutionary algorithms (EAs) (Holland, 1975 and Goldberg, 1989) have been shown to be useful in tasks involving a large search area, including searching for OWA weights (Nettleton and Torra, 2001). In the research presented in this paper, we will employ EAs to search for appropriate weights for WAs and OWAs for use in a security assessment problem.

The data set used is from a decision making exercise that was conducted at GCHQ, Cheltenham, UK, the UK government's signals intelligence and information assurance agency. Thirty nine GCHQ selected cyber-security experts including system and software architects, security consultants, penetration testers, vulnerability researchers and specialist systems evaluators, took part in two survey exercises. In the first, they were asked to rank a set of ten technical attacks in order of how difficult they would be to complete without being noticed. The set included attacks via a voice over IP (VOIP) client, a malformed document via email and a malicious website. In the second exercise, experts were asked to rate the difficulty of compromising/bypassing the twenty six security components that make up the ten attacks from the previous exercise. Security components included anti-virus software, cryptographic devices and firewalls. They were also asked to rate specific characteristics of each component, for example, the complexity of a component or the public availability of tools that could be used by a potential attacker to compromise/bypass the component. Undertaking this type of survey can identify particularly weak points in a system, and thus potential "breachpoints" in the system. As such, this activity is an important



Fig. B.1 - Structure of assessment data.

part of the security assessment process. The result is a data set containing three levels of assessment:

- 1. Rankings of technical attacks on a proposed system.
- 2. Ratings of the level of security offered by security components in the proposed system.
- 3. Ratings of specific characteristics of security components in the proposed system.

Fig. B.1 shows the structure of the data set. With this data set we will show how, using WAs and OWAs with EAs, the following can be achieved:

- 1. Ratings of the difficulty of attacking and evading security components using ratings of specific characteristics.
- 2. Indication of the relative contribution each characteristic makes to the overall difficulty rating for a security component.
- 3. Rankings of specific technical attacks using ratings of security components.
- 4. Indication of the relative contribution components make to the attack ranking based on their difficulty rating.
- A combination of the previous aggregations, in which rankings of specific technical attacks are computed using security component ratings that have been created using ratings of specific characteristics.

As we have data at all three levels, we are able to compare the derived security component ratings and attack rankings with experts' actual ratings/rankings to validate them. Potentially, this method could be used with a database of existing ratings for components/characteristics to aid in security assessments on proposed systems, reducing load on the cybersecurity experts who currently carry out such assessments.

The paper is structured as follows. Section 2 provides an overview of the problem of performing security assessments on information systems, a review of machine learning approaches to this problem, WA/OWA operators and EAs. Section 3 describes the decision making exercise conducted with cybersecurity experts at GCHQ, while Section 4 details how EAs, OWAs and WAs have been implemented for this study. In Section 5, overall difficulty ratings for security components are produced using ratings of specific characteristics of each component. Moreover, we also produce weightings that signify the contribution each characteristic makes to the overall difficulty rating. Then, in Section 6 rankings for specific technical attacks are produced using difficulty ratings of their constituent components, along with weightings that give an indication of which components contribute most to the rankings. Section 7 combines the processes of the previous two sections to derive ratings and rankings for specific technical attacks from ratings of specific characteristics of their constituent components. Finally, the outcomes of the work are discussed in Section 8, and Section 9 summarises the contributions of the research and considers future directions of research.

2. Background

2.1. Cyber-security assessment

In current times, the use of the Internet and cellular networks has become the norm for a variety of every-day tasks including banking, e-mail, e-government, e-commerce, VOIP, social networking and mobile telecommunications. An everincreasing number of sensitive interactions are taking place online, making them susceptible to attack by cyber-criminals who may read, modify or delete sensitive data. Concurrently, the volume and availability of tools to aid would-be attackers in their acts are also increasing. These tools lower the barrier to entry for attackers, as far less technical expertise is required to operate them than to launch an attack from scratch. Exacerbating the situation further is the fact that it is often unclear exactly how defensive security products achieve their claims, and indeed, whether they are effective at all.

With the increasing threat of cyber-crime the practice of carrying out security assessments on proposed systems has become a critical part of the systems design process. Understanding the threats that attackers pose to a proposed system, their consequences, how those threats can be addressed and what represents an appropriate level of security is a complex problem involving many factors. These factors include (Tregear, 2001):

- Data are limited on certain threats, such as the likelihood of a sophisticated hacker attack and the costs of damage, loss, or disruption caused by events that exploit security weaknesses.
- 2. Some costs, such as loss of customer confidence or disclosure of sensitive information, are inherently difficult to quantify.
- 3. Although the cost of the hardware and software needed to strengthen controls may be known, it is often not possible to precisely estimate the related indirect costs, such as the possible loss of productivity that may result when new controls are implemented.
- Even if precise information were available, it would soon be out of date due to fast paced changes in technology and factors such as improvements in tools available to would-be intruders.

The consequence of these difficulties is that it is necessary for highly experienced cyber-security experts to carry out assessments of system designs including their hardware/ devices, architecture, security software and practices before they are implemented. The goal of such assessments is to ensure that there is a proportionate level of security offered by the proposed system in line with the consequences of a successful attack. Clearly, investing too little in security measures leaves a system exposed to an unacceptable level of risk; however, investing too much is also a problem as it involves expending an unnecessary amount of money and effort on security when the consequences of a successful attack are considered.

Unfortunately, it is often the case that there is insufficient expertise to carry out all of the security assessments necessary. The ever increasing number of information systems, which themselves are of increasing complexity and exposed to constantly evolving threats, mean that a greater number of security assessments are required more quickly than ever before. In this research we focus on modelling parts of the assessment process in order to create decision aids that can be used by experts to reduce load, and by non-experts to achieve an approximation of security levels for security components and specific technical attacks.

2.2. Machine learning approaches to security assessment

Over the years there have been many machine learning approaches to computer security, ranging from classic algorithm based studies (Lane, 1998) to immune system inspired approaches (Kim et al., 2007). What almost all of these past approaches have in common is that they attack the problem from a quantitative point of view, based on evaluating datasets such as network traffic, but without an attempt to integrate often highly valuable knowledge held by security experts. Examples include what is probably the earliest use of fuzzy sets in this context (Clements, 1977), intrusion detection based on multisensor data fusion Bass (2000) or fuzzy cognitive maps Siraj et al. (2001), and various automated decision systems to support risk or threat analysis (Dondo, 2007; Linkov et al., 2006; McGill and Ayyub, 2007; Ngai and Wat, 2005; Shah, 2003; Sun et al., 2006; Van de Walle and Rutkowski, 2006). Other approaches include anomaly detection algorithms, e.g. using Bayesian approaches (Androutsopoulos et al., 2000) to detect spam in emails, or self-organising map algorithms to learn combinations of external signals and system call IDs to detect unusual patterns (Feyereisl and Aickelin, 2012). Similar to anomaly detection is the area of intrusion detection, which again typically relies on unsupervised machine learning techniques for two class classification approaches of "normal" and "abnormal" behaviour, for example the work by Twycross and Aickelin (2010).

More general approaches to systematising and supporting methods for the design of secure systems have been reviewed by Baskerville (1993), Dhillon and Backhouse (2001) and more recently by Jansen and Gallagher (2009). As a refinement of more general approaches, this paper focuses on extracting and making use of and leveraging the insight held by cyber security experts. Experts are commonly good at assessing the security of parts of systems – such as rating the vulnerability of individual hops or attack paths. However, the major challenges for experts are:

• The large number of hops and attack vectors in systems, i.e. the finite number of suitably qualified experts makes the timely assessment of hops and attack paths highly challenging for many users (e.g. companies).

• The aggregation of often different ratings for the same components by different experts, e.g. based on different expert backgrounds or levels of expertise.

In this context, our paper focuses on alleviating the task load for available experts by introducing, demonstrating and evaluating a novel approach to partially automating the rating of attack vectors based solely on individual hop ratings. The approach proposed employs linear order statistics, specifically the ordered weighted average (OWA), to fuse individual hop ratings into an overall attack path assessment, while the weights of the OWA are determined using an evolutionary algorithm, resulting in overall high quality attack vector vulnerability assessments which closely follow those of experts (if these experts directly assess the vectors). The closest body of works in computer security that is related to our approaches is that of attack graphs and scenario creation, e.g. the work by Tedesco and Aickelin (2008). However, in these approaches the aim tends to be to develop improved techniques to construct and rate attack graphs, rather than facilitating and partially automating the overall process of integrating expert knowledge in cyber security assessments.

2.3. Weighted and Ordered Weighted Averages

In the experiments shown in this paper, two aggregation methods are employed: weighted averages (WAs), and ordered weighted averages (OWAs) (Yager, 1988). OWA operators were chosen for the task of aggregating security component ratings to produce attack rankings following discussions with a group of GCHQ technical experts. In our discussions the hypothesis emerged that the difficulty of a particular attack is largely determined by the most difficult security component to attack or evade; the remaining components contribute to the difficulty in proportion with their own difficulty. Initially, we tried simple maximum operators to see whether they could be used. In practice however, this results in a lot of attacks having tied ranks, providing little insight. This is because, if a particularly difficult component appears in multiple attacks, they will all receive the same rating. OWA operators allow us to assign more weight to the most difficult components, while accounting for the difficulty of the remaining components too. This greatly reduces the potential for tied ranks, and creates meaningful rankings that distinguish between attacks.

Let X be a set of N information sources (e.g., reviewers, experts, etc.) with each information source contributing evidence x_i , i = 1...N. The standard weighted average, WA, combines the information from all the sources by associating the evidence from each source with a given weight. More formally, consider a set W of weights w_i corresponding to each source x_i , where $\sum w_i = 1$. Then:

$$WA(X) = \sum_{i=1}^{N} w_i x_i$$
⁽¹⁾

The ordered weighted Average, OWA, combines the information from all the sources, ordered by the size of the evidence (largest to smallest), using a pre-defined vector of weights, W'. More formally, consider the vector of evidence $O = (o_1, ..., o_N)$, formed by the ordering (largest to smallest) of the elements in X, such that $o_1 \ge o_2 \ge ... o_j$. Then:

$$OWA(X) = \sum_{j=1}^{N} w'_{j} o_{j}$$
⁽²⁾

Whereas in the WA, the weights are associated with each source, in the OWA, the weights are associated with an ordered set of the evidence by all sources. Thus, in the OWA, a change in contributed evidence can result in a different ordering and thus in a different mapping of weights to evidence. For example, in this work we assign a weight to the most difficult component, the second most difficult component and so on. Each of the weights is multiplied by the corresponding component, the first weight is multiplied by the first and thus largest component and so on. If the first weight is near to one, the OWA behaves similarly to a "maximum" operator, thus resulting in the overall assessment being based nearly exclusively on the most difficult component.

OWAs are commonly used in decision making problems to create aggregate scores and/or ratings. In Canós and Liern (2008), OWA operators are used in a personnel selection problem to aggregate selection criteria. In Badea et al. (2011) an OWA is used to compute a rating of the security of energy supply, Sadiq et al. (2010) describes the use of an OWA to aggregate performance indicators of small water utilities creating an overall performance assessment. Imamverdiev and Derakshande (2011) show how a fuzzy OWA operator can be used for rating information security products in terms of reduction of information security risk, and in Merigó and Gil-Lafuente (2010) and Merigó and Gil-Lafuente (2011) modified OWA operators are applied to a financial product selection problem and a football team player selection problem respectively.

2.4. Evolutionary Algorithms

Evolutionary algorithms (EAs) are a set of well used heuristic search methods that mimic aspects of biological evolution to *evolve* solutions. The most popular EA, the genetic algorithm (GA) (see Holland, 1975 and Goldberg, 1989), begins with the generation of a random initial population of solutions to which fitness-based *selection and copy*, *crossover* and *mutation* operators are applied in order to discover near-optimal solutions. The advantage of the GA is that it is able to search a large area of the solution space, while evaluating a small proportion of the possible solutions. This is critical for problems like those seen in this work in which there are far more possible solutions than it is practical to evaluate. The use of a GA to search for suitable OWA weights is demonstrated in Nettleton and Torra (2001) and Torra (2011), and in our own previous work (Miller et al., 2013a).

The work in the current paper significantly extends the work seen in Miller et al. (2013a) in which EAs were employed to search for appropriate OWA weights to be used for aggregating ratings of security components to produce rankings of technical attacks, and to search for WA weights to be used to aggregate ratings of aspects of security components to produce overall security component ratings. Using an EA in this way not only allows us to discover suitable aggregation operators, it also gives an indication of the relative importance of each component of the aggregation. For example, by looking at the resulting OWA weights, we will be able to confirm or contradict our hypothesis that the most difficult to attack/evade security component contributes the most to the attack ranking. As the EA could theoretically arrive at any OWA, and is led purely by the fitness of an OWA (i.e., how well the resulting rankings match individuals' actual rankings), the OWAs produced are valid guides to the relative contribution made by each security component. Similarly, the WA weights found will provide insight into which characteristics contribute most to the overall rating, and which have little effect on the overall difficulty of attacking/evading a security component.

2.5. Statistical analysis

To make comparisons between the results produced by the OWA/WA operators and the experts' actual rankings, Spearman's Rho is used. Spearman's Rho (Spearman, 1904), also called Spearman's rank correlation coefficient, measures the statistical dependence of two sets of rankings (ordinal data). The coefficient is a number in [-1, 1] that indicates the level of correlation; 1 denotes a perfect positive correlation, 0 means there is no correlation, and -1 denotes a perfect negative correlation. To measure the correlation between component characteristics and overall difficulty questions Pearson's r (Pearson, 1895) is used. Pearson's r, also called the Pearson product-moment correlation coefficient (PPMCC), is a method that measures linear correlation between two sets of values (interval or ratio data); a value of zero indicates no correlation and a value of one indicates perfect correlation. For both methods, generally, values above 0.5 are interpreted as strong correlation.

2.6. Experimental validation

The experimental results and statistical analyses presented in this paper are not tested in a formal manner against, for example, pre-specified null hypotheses or similar. Thus, in statistical terms, it should be viewed as exploratory data analysis rather than as formal hypotheses testing (sometimes called confirmatory data analysis), as discussed by Tukey (1962). Overall validation of the experimental methods in this paper is presented in terms of the degree to which it is possible to model parts of cyber-security decision makers processes using statistical aggregation of components within attack vectors. As this is exploratory, there are no explicit tests for falsifiability; however, there is an informal null hypothesis that security "experts" are not actually experts, and that no correlations, associations or aggregations could be found that would link rating of sub-components to rankings of attack vectors. But, the main purpose of this work is to carry out exploratory data analysis in this context to explore to what degree it may be possible to submit security assessment to more systematic methods, which in future could lead to new hypotheses with associated methods of new data collection and analysis.

3. Data elicitation

In this section we will describe the data elicitation exercise conducted at GCHQ, Cheltenham, UK. As the data and actual elicitation are vital to the design and functionality of the automated reasoning system introduced later, we proceed by describing the elicitation process and resulting data in detail. Thirty nine highly experienced GCHQ selected cyber security experts from government and commercial backgrounds including system and software architects, technical security consultants, penetration testers, vulnerability researchers and specialist systems evaluators took part in a decision making exercise. While the detail of the actual attack scenarios and security components is not available for general publication, we provide the actual numeric "anonymised" data in the paper.

3.1. Scenario

For the purposes of the exercise, a scenario was created and presented by senior technical staff at GCHQ. The scenario was designed to be a typical example of a government system. The system has a range of core services and back end office facilities, along with remote sites and mobile access. Fig. B.2 provides an overview of the system architecture. The most sensitive information is held in core systems, with assets rated at Business Impact Level 3 (BIL3). Business Impact Levels are the UK government's standard classification scheme for rating the ramifications of assets being disseminated, altered or destroyed, ranging from no consequences (BIL0) to catastrophic consequences (BIL6).

The experts were presented with information regarding the architecture of the system and its components, and allowed to ask questions. In real-world assessments, a great deal of information about a system and its components is required in order to perform a security assessment, including details of hardware and software, version/revision numbers and the frequency with which updates and patches are applied. In order to address this in the scenario, without overcomplicating the exercise (making it infeasible), the experts were asked to make assumptions about the system, specifically, that the software/ hardware used was typical of this type of government system and that security policy was being applied in manner consistent with how they typically find it in their day-to-day work. This was acceptable for the experts, as they all have considerable experience with the type of BIL3 government system presented in the scenario, and how security policies are generally applied.

In addition to details of the system, the experts were also given a list of 10 technical attacks on the system, and the 26 security components that make up the attacks. Fig. B.3 shows an example of the information the experts were given about the technical attacks. In this attack an email with a malicious attachment is sent to a recipient within the system, when opened the attachment runs an exploit taking control of the recipient's PC, allowing the attacker to launch further attacks from their machine. Fig. B.3a shows the path the attacker takes from outside the system, through the back office systems. Once this is achieved the attacker proceeds to the mobile sites as shown in Fig. B.3b, completing the attack by compromising a client desktop.

The five steps in the attack are labelled A to E in Fig. B.3, Table B.1 provides details of the security components involved. Each of these components must be attacked (compromised) or evaded (bypassed) in order for the attack to be successful. After being shown the 10 attacks and their



Fig. B.2 - Scenario system architecture. Note that symbols are self-explanatory following standard computer network diagram drawing guidelines, based on Cisco icons (see

http://www.cisco.com/c/en/us/about/brand-center/network-topology-icons.html).

constituent components, the experts were asked to complete two survey exercises:

- 1. Rank the 10 technical attacks in order of difficulty to complete without being noticed.
- 2. Rate the 26 security components in terms of:
 - (a) The overall difficulty of attacking or evading the component.
- (b) Individual factors that contribute to the difficulty of attacking or evading the component.

Each activity was conducted in silence, with experts sat at separate desks, to avoid experts influencing each others' opinions. Ranking the technical attacks like this is an intuitive way of establishing the relative difficulty of completing each of the attacks successfully. Experts make direct



comparisons between attacks. e.g., "Is attack x more or less difficult than attack y?", in order to arrive at a ranked list. This approach can make it easier for experts to identify salient attacks, as in some cases it is difficult to place a precise value on the difficulty of an attack. When rating security components, experts were asked to give a rating on a scale from 0 to 100 to the question "Overall, how difficult would it be for an attacker to successfully attack/ evade this component without being noticed?", and a series of questions about factors that contribute to the overall

Table B.1 – Components in example attack.				
Step	Attack/ evade	Component		
А	Evade	Content checker		
В	Evade	Anti-virus software		
С	Attack	PDF renderer		
D	Evade	Anti-virus software		
E	Attack	Client access controls		

difficulty of attacking/evading a component. These questions were created by GCHQ technical experts, who identified what they believed to be the important factors that contribute to the difficulty of attacking/evading security components. The rating scale from 0 to 100 was completely arbitrary, with no absolute meaning of 0 or 100 being conveyed to the experts, and the response included both a perceived location ("mean") and uncertainty ("variance") of opinion as described in more detail in Miller et al. (2013b).

Two sets of questions were used, one for when an attack required a component to be attacked (compromised), and another for when an attack required a component only to be evaded (bypassed). The categorisation of components into "attack" or "evade" was made by the GCHQ technical team who designed the scenario. While it is possible that some components may in practice be either attacked or evaded, and that different attach vectors may be possible, such alternatives were not considered. That is, the list of attack vectors created, and whether the components were compromised or bypassed within such an attack, were designed and specified by the GCHQ technical team.

The questions asked when a component needed to be attacked were:

- How complex is the target component (e.g., in terms of size of code, number of sub-components)?
- How much does the target component process/interact with any data input?
- How often would you say this type of attack is reported in the public domain?
- How likely is it that there will be a publicly available tool that could help with this attack?
- How inherently difficult is this type of attack? (i.e., how technically demanding would it be to do from scratch, with no toolset to help.)
- How mature is this type of technology?
- How easy is it to carry this attack out without being noticed?
- Overall, how difficult would it be for an attacker to do this?

For components that needed only to be evaded, the following questions were asked:

- How complex is the job of providing this kind of defence?
- How likely is it that there will be publicly available information that could help with evading this defence?
- How mature is this type of technology?
- Overall, how difficult would it be for an attacker to do this?

This type of questioning allows experts to give an indication of the absolute difference in difficulty between security components, unlike ranking which provides an intuitive way to obtain an ordering but elicits no information about the degree of difference between attacks.

The dataset created by this exercise is made up of three levels:

- 1. Rankings of attacks.
- 2. Ratings of the overall difficulty of attacking/evading specific security components that make up attacks.
- Ratings of characteristics/factors that contribute to the overall difficulty of attacking/evading specific security components.

Details of analysis work carried out previously on the resulting data can be found in Miller et al. (2013b).

The exercise and methods described in this paper represent an important part of performing a security assessment on a proposed system. The process of rating security components and highlighting the easiest ways to attack a system via its weakest components is critical to understanding the levels of risk posed by a system design. In this study we will show how rankings of attacks can be produced from overall ratings of components, how overall ratings of components can be created from ratings of characteristics of components, and how the two stages can be chained together to produce attack rankings from ratings of characteristics of components.

4. Data analysis methods

4.1. OWA/WA aggregation

The methodology in this paper is based on determining the "optimum" weights for either WA or OWA aggregation of the assessment of difficulty of compromising the various components in a range of attack vectors. This in turn, enables both the ranking of attack vectors in terms of their overall difficulty as well as providing an understanding of exactly which components in a security system contribute the most/least to an overall system being compromised. Below, we briefly summarise the methodology and detailed rationale for both the WA and the OWA.

In all, 26 security components were rated, each belongs to one or more of the 10 specified technical attacks. Table B.2 lists the attacks, and their constituent security components. Notice that some components appear more than once in the same attack, this is because in some cases more than one instance of a component needs to be attacked or evaded for an attack to be completed successfully. For example, an attacker may have to compromise multiple firewalls in order to obtain access to their target. The order of components is important – alternative orderings, with perhaps alternative attack methodologies, may be possible in the real world, but these would be considered as different overall attacks.

With the information regarding the composition of attacks, a WA can be employed for the aggregation of the ratings of the difficulty of attacking/evading a security component, to produce an overall difficulty rating. In the instance of the WA, weights

Table B.2 – Attacks and their constituent security components.				
Attack	Components			
1	2,3,1,4,5			
2	6,7,6,8,4			
3	9			
4	10,11,4,5			
5	12,13,2,3,14,15,4,5			
6	16,16,17,4,5			
7	6,18,4,5			
8	19,20,21			
9	22,23,24			
10	25,26,1,4,5			

are applied to specific factors, e.g., the complexity of a component. The result is a set of derived overall difficulty ratings for all 26 security components. The core challenge here is the search for the "optimal" weights of the given WA. To address this, this paper proposes the use of evolutionary algorithms (EAs) which will be discussed in the next section.

Similarly, an OWA is used to produce attack rankings from overall difficulty assessments of their constituent hops. However, as noted in Section 2.3, the OWA matches a vector of weights to the ordered set of evidence (difficulties) for the individual components. For instance, an OWA can be used to aggregate overall difficulty ratings of components to produce a rating for each of the ten attacks. As an example, Table B.3 shows the overall difficulty ratings that an expert gave for attacking/evading the security components in Attack 1.

If we want to aggregate these values into an overall rating for Attack 1 that gives more weight to the most difficult components, we could choose the weights

W' = (0.33, 0.27, 0.20, 0.13, 0.07).

The next step is to order the ratings

 $X = \{25.00, 40.00, 20.50, 40.00, 70.00\}$

by difficulty to create the ordering

O = (70.00, 40.00, 40.00, 25.00, 20.50).

Equation 2 shows how the OWA can then be applied by multiplying each weight by its corresponding element in the ordering. In this case the resulting value is 46.59. When we compare this with the mean value 39.10, it can be seen that this OWA operator has given more weight to those compo-

Table B.3 – Overall difficulty ratings for the components in attack 1.	security
Component	Overall difficulty
1	25.00
2	40.00
3	20.50
4	40.00
5	70.00

nents with higher difficulty ratings. If we then repeated this process with the remaining nine attacks, a complete set of attack ratings would be produced that could be used to generate attack rankings.

4.2. EA implementation

In this work, EAs (as introduced in Section 2.4) are used to optimise the weights for:

- 1. WAs, used to aggregate ratings of difficulty of factors to produce the overall difficulty ratings of security components;
- OWAs, used to aggregate the overall difficulty ratings of security components to create rankings of the overall attack vectors.

The EA has been implemented for both the WA and OWA cases as follows.

Solutions are represented as a vector of weights. In this example, the maximum number of security components in an attack is eight; therefore, each vector contains eight weights. Those attacks that do not have eight components are padded out with zeros. This is to avoid concentrating weightings (which must add up to 1) on fewer components, which can have the effect of making an attack more difficult purely because the weights are concentrated on fewer component ratings. In general, an attack should get more difficult as components are added to it.

An initial population of individuals is created at the beginning of the algorithm. Each individual is produced by generating seven random points on a line from 0 to 1. From this, eight weights are created by taking the distances between 0, each of the seven points, and 1. This ensures that the result is eight values that add up to 1.

To compute fitness each expert's actual attack ranking is compared with the attack ranking derived from the overall security component rating with the current OWA. The comparison is made using Spearman's rank correlation coefficient, which produces a value between -1 (perfect negative correlation), 0 (no correlation) and 1 (perfect positive correlation). An error value is then calculated by subtracting the correlation coefficient from 1, so a perfect positive correlation produces an error of 0, and anything less produces an error value greater than 0. Once an error value has been calculated for each expert, the mean squared error (MSE) is computed to give an error value for each solution in the population.

Selection is achieved by sorting the population of solutions by fitness, and then generating the indexes of selected parents from a complementary cumulative distribution. A lower numbered individual is more likely to be chosen, ensuring that fitter individuals a more likely to be selected, though potentially any individual can be selected.

To guarantee that the fittest individual in successive generations is not worse that in preceding generations *elitism* selects the best individuals from the current generation and copies them unaltered into the next generation. A similar operator is the copy operator, which takes a parent picked using the described selection method and copies them into the next generation. The *mutation* operator used in this EA implementation randomly selects two weights from a solution, increases one by a small amount and reduces the other by the same amount. This ensures that the weights still add up to one afterwards. The resulting weights are validated, if either becomes greater than 1 or less than 0, another two elements are selected.

The crossover method used is a single point crossover. Two parents are selected and a child is created that consists of the first four weights of the first parent, and the last four from the second parent. It is unlikely at this point that the weights will add up to 1, so they are normalised. This method of crossover ensures valid OWAs, while preserving the characteristics of each contributing parent.

Finally, the EA's termination criteria are set to be a specific number of generations, after which the algorithm stops.

5. From factor ratings to security component ratings

In Section 4, we described how an EA has been implemented for use in this research. In this section we employ the described EA to perform a search for WA weights. In these experiments, we show how overall difficulty ratings for security components can be obtained from ratings of their characteristics using WAs where the weights have been discovered using an EA. As well as producing overall difficulty ratings, this process will also highlight which questions contribute most/least to the overall ratings. In this case it is not rankings that are being compared, it is ratings. Because of this, Spearman's rho is not an appropriate method for comparison so an alternative method of calculating fitness was established. The "error" is directly calculated by taking the sum of absolute differences between the derived hop ratings and the experts' actual hop ratings. Each expert's error value is then used to calculate a MSE value for all experts, which is used as the fitness value for each solution.

When performing data analysis with the ratings of characteristics/factors we found that the relationship between them and the overall difficulty of attacking/evading a security component varies between question sets. This is because they contain different numbers of questions, and some questions are specific to a question set. Because of this, we will perform separate experiments with the questions for attacking and evading security components, finding WAs specific to each.

5.1. EA configuration

Initially, a series of tests were conducted to find ideal configurations of the EA. Full details of these experiments and their results can be found in Appendix B. The best results for each question set can be found in Table B.4; the best EA configurations are shown in Table B.5. The best EA found for each question set is used in the next subsection for extended experiments.

5.2. Extended experiments

In these extended experiments the best EA configuration found for each question set will be used in 30 runs with differing

Table B.4 – EA-WA: EA configuration – best results.				
Test	Mean Sp.	MSE		
Attack				
15	0.7931	194.0125		
Evade				
1	0.2270	420.2156		

Table B.5 – EA-WA: EA configuration – best EAs.						
Attack						
Test	Gens	Рор	Сору	Cross	Mut	
15	400	155	0.20	0.20	0.59	
Evade						
Test	Gens	Рор	Сору	Cross	Mut	
1	250	250	0.00	0.20	0.79	

random seeds. The purpose of these experiments is to assess the consistency of the EAs, and to see the best results we can reasonable expect with the EA. In experiments with the attack question set the best configuration (Test 15) will be used; Appendix B shows that multiple configurations achieved the best result for the evade question set, of these we have arbitrarily chosen to use the configuration from Test 1. Table B.6 shows a summary of the results for the attack question set. The results are extremely stable, there is little variation in the results over the 30 runs. Table B.7 gives details of the best WA found.

The results of the experiments with the attack question set show that there are clear differences in the relative importance of each question in deriving the overall difficulty of a security component. In order, the questions are ranked as shown in Table B.8. For comparison, the mean (over all 39 experts) Pearson's r for the correlation between the ratings given for each characteristic, and the overall difficulty is also given. The weightings shown are taken from the test that resulted

Table B.6 – EA-WA: extended experiments – attack.					
Spearman	Spearman's Rho				
Max	Min	Mean	Standard deviation		
0.7940	0.7919	0.7935	0.0005		
MSE					
Max	Min	Mean	Standard deviation		
194.2998	193.9896	194.0856	0.0741		

Table B.7 – EA-WA: extended experiments – best WA – attack.						
Best wei	ghts					
Test	Complexity	Interaction	Frequency			
22	8.43E-05	0.0012	0.1622			
Tool	Inherent	Maturity	Unnoticed			

0.0438

0.1719

0.2189

0.4020

Table B.8 – EA-WA: extended experiments – question ranking – attack.

Rank	Question	Weight	Mean correlation
1	How inherently difficult is this type of attack?	0.4020	0.6454
2	How likely is it that there will be a publicly available tool that could help with this attack?	0.2189	0.6732
3	How easy is it to carry this attack out without being noticed?	0.1719	0.4543
4	How often would you say this type of attack is reported in the public domain?	0.1622	-0.6268
5	How mature is this type of technology?	0.0438	0.1701
6	How much does the target component process/interact with any data input?	0.0012	-0.2741
7	How complex is the target component?	8.43E-05	-0.0623

in the lowest MSE (Test 22), though all tests produced similar results.

The inherent difficulty rating has the highest weighting, and it is also strongly correlated with the experts' overall ratings for security components. This may be because the experts do not make a strong distinction between the inherent difficulty of attacking a security component and the overall difficulty, as the terms are similar. The weights attributed to the ratings regarding factors of component difficulty are in line with the correlation between experts' ratings of each factor and their overall rating for each component. In view of the weightings and correlations, it is reasonable to say that the questions ranked 1 to 4 have a demonstrable relationship with how experts rate the overall difficulty of compromising a security component. There does not appear to be a significant relationship between how experts rated the factors in questions 5 to 7 and how they rated the overall difficulty of compromising security components.

From this we could infer that these three questions about the security components are not useful when trying to determine the difficulty of compromising a component, and that only the four with high correlation/weighting are necessary. Alternatively, it could be that the four high correlation/ weighting questions were the only ones that experts are able interpret and answer in a consistent manner with the information given. Regardless, using the four questions overall ratings that are closely correlated with participants' actual ratings of overall difficulty can be produced.

Table B.9 provides a summary of the results with the evade question set. Like the experiments with the attack question set, the results are extremely stable. Over 30 runs, 28 produced the same MSE. Table B.10 shows the WA found in Test 1, though all weightings are within 0.0003 of those shown.

Again, some questions are consistently given more weight than others. Table B.11 provides a list of the difficulty factor questions ranked by the weight they were given by the EA. The Mean correlation column in Table B.11 gives the mean Pearson's r value for the correlation between experts' ratings of

Table B.9 – EA-WA: extended experiments – evade.					
Spearman's Rho					
Max	Min	Mean	Standard deviation		
0.2273	0.2270	0.2270	5.38E-05		
MSE					
Max	Min	Mean	Standard deviation		
420.2158	420.2156	420.2156	4.76E-05		

Table B.10 – EA-WA: extended experiments – best WA – evade.				
Test	Best weights			
	Complexity	Information	Maturity	
1	0.2185	0.4971	0.2845	

factors and their overall ratings. When compared with the correlation values for the attack questions, these are much lower, which may explain why the EA is unable to find a WA that produces overall ratings that are strongly correlated with experts' actual overall ratings.

6. From security component ratings to attack rankings

In this section, we describe how the EA was employed to perform a search for OWA weights. The OWAs are to be used for aggregating overall difficulty ratings of security components to produce difficulty rankings of attacks. In these experiments three groupings of experts have been used: *odd* experts i.e., experts assigned an odd number, *even* experts i.e., experts assigned an even number and *all* experts. This has been done so that OWAs discovered with one group (e.g., *odd*) can be applied to the alternate group (e.g., *even*) to assess their robustness, and the *all* group can be used to obtain the best OWAs that can be found for the entire group of experts for comparison purposes. Extended experiments are used with the best configuration found with the *all* group to test the stability of the configuration, and to further explore the best OWAs that can be found.

Table B.11 – EA-WA: extended experiments – question ranking – evade.					
Rank	Question	Weight	Mean correlation		
1	How likely is it that there will be publicly available information that could help with evading defence?	0.4971	-0.3307		
2	How mature is this type of technology?	0.2845	-0.0348		
3	How complex is the job of providing this kind of defence?	0.2185	0.0278		

Table B.12 – EA-OWA: EA configuration – best OWAs.								
Test		Best weights						
	1	2	3	4	5	6	7	8
Even								
12	0.8907	0.0011	0.0323	0.0120	0.0070	0.0174	0.0035	0.0359
13	0.8899	0.0044	0.0297	0.0126	0.0078	0.0242	0.0030	0.0284
Odd								
13	0.7858	0.0223	0.1285	0.0471	0.0115	0.0031	0.0015	0.0002
16	0.7854	0.0246	0.1248	0.0479	0.0117	0.0033	0.0001	0.0021
All								
1	0.9582	0.0028	0.0242	0.0003	0.0057	0.0029	0.0002	0.0056
2	0.9621	0.0017	0.0221	0.0003	0.0045	0.0059	0.0009	0.0025

6.1. EA configuration

Initially, a series of experiments was conducted to find ideal configurations of the EA for use in these tests. Full details of these experiments and their results can be found in Appendix A.1. The best OWAs found are shown in Table B.12 and the best results from these experiments are shown in Table B.13. The best OWAs found for the *odd* and *even* groups will be assessed for their robustness, and the best EA configuration for the *all* group will be used in extended experiments.

6.2. OWA robustness

In order to assess the robustness of the OWAs found by the EA, the best OWAs found for the *odd* group were applied to the *even* group and vice versa. The results of these experiments are provided in Table B.14. It can be seen that while there is some degradation of performance when the OWAs are applied to the alternate group, they produce rankings that are strongly correlated with experts' actual rankings (i.e., Spearman's Rho >0.5 indicating strong correlation) suggesting that they are robust. As these OWAs work well on their alternate (unseen)

Table B.13 – EA-OWA: EA configuration – best results.				
Test	Mean sp.	MSE		
Even				
12	0.6885	0.1421		
13	0.6866	0.1421		
Odd				
13	0.5732	0.2519		
16	0.5733	0.2519		
All				
1	0.6159	0.2069		
2	0.6159	0.2069		

Table B.14 – EA-OWA: OWA robustness experiments.				
Test	Mean sp.	MSE		
Best even OWA applied to odd				
12	0.5228	0.3013		
13	0.5146	0.3118		
Best odd OWA applied to even				
13	0.5915	0.2381		
16	0.5947	0.2337		

group, it is reasonable to expect that OWAs trained with sample data will work well on unseen data.

6.3. Extended experiments

The final set of experiments take the best configuration found with the all grouping and conduct 30 runs with varying random seeds to allow assessment of the stability of the EAs, and to give a better picture of the best OWAs that can reasonably be expected using the proposed approach. The EA configuration used is shown in Table B.15. A summary of the results of the extended experiments can be found in Table B.16. The summary shows that with 30 different random seeds, the results are very stable, there is minimal difference between the solutions found and they all result in OWAs that produce rankings that are strongly correlated with experts' actual rankings. Also of interest is that in all 30 runs the solutions found never place less than 0.92 of the weight on the most difficult to attack/evade security component, corroborating our hypothesis that this is the most important component when assessing the difficulty of an attack.

7. From factor ratings to attack rankings

In this section we will use the best WAs and OWAs, identified as described in the previous sections, to derive attack rankings

Table B.15 – EA-OWA: extended experiments – EA configuration.					
Gens	Рор	Сору	Cross	Mut	
250	250	0.00	0.20	0.79	

Table B.16 – EA-OWA: extended experiments.				
Spearman	Spearman's Rho			
Max	Min	Mean	Standard deviation	
0.6200	0.6104	0.6150	0.0022	
MSE				
Max	Min	Mean	Standard deviation	
0.2137	0.2057	0.2092	0.0020	

Table B.17 – Factor ratings to attack rankings: WAs and OWA.			
Index		WA	
	Attack	Evade	OWA
1	8.43E-05	0.2185	0.9484
2	0.0012	0.4971	6.7E-05
3	0.1622	0.2845	0.0368
4	0.2189	-	0.0120
5	0.4020	-	0.0008
6	0.0438	-	0.0016
7	0.1719	-	0.0002
8	-	-	0.0002

from the difficulty factor ratings. This is a two stage process: firstly, the best WAs will be used to compute overall difficulty ratings for security components from ratings of factors of that difficulty; secondly, the derived overall difficulty ratings will be aggregated using the best OWA to create attack rankings. Table B.17 shows the WA and OWA operators used, and Table B.18 gives a summary of the results. Note that in this case pairs of rankings are being compared, as such, error is calculated by subtracting the Spearman's Rho value from 1, as in Section 6. Because of this it can be seen that, for example, the minimum error is the complement of the maximum Spearman's Rho as they refer to the same individual. In the previous experiments where overall difficulty ratings for security components were being compared, error was calculated by taking the absolute value of the differences between the actual ratings provided by the experts, and the ratings derived from the difficulty factor ratings.

The mean Spearman's Rho is 0.5562; this indicates that there is a strong correlation between the rankings produced using the difficulty factor ratings and those that the experts actually gave. It is clear that most experts answer the questions relating to the factors of difficulty in such a way that it is possible to create overall ratings for security components and rankings of attacks that closely match their opinions.

8. Discussion

The proposed method demonstrates that it is possible to model parts of cyber-security decision makers processes using WAs and OWAs.

For our first set of experiments we used an EA to search for WA weights that can be used to take experts' ratings of factors that contribute to the difficulty of attacking/evading a secu-

Table B.18 – Factor ratings to attack rankings: results.				
Spearmai	Spearman's Rho			
Max	Min	Mean	Standard deviation	
0.9636	-0.4303	0.5562	0.3387	
Error				
Max	Min	MSE	Standard deviation	
1.4303	0.0364	0.3117	0.3387	

rity component, and produce overall difficulty ratings for those components. The results showed that for the attack question set it is possible to produce overall ratings that closely match experts' actual rankings. With the evade question set however, much poorer results were obtained. The EA highlighted the factors that contribute the most to the overall rating, specifically questions regarding the availability of tools to help with an attack, the difficulty of achieving the attack unnoticed, the frequency of such attacks being reported and the inherent difficulty of completing an attack. The correlation of experts' answers to these questions and their overall difficulty ratings show that they have moderate/strong correlation, further supporting their importance for this type of assessment. Conversely, questions regarding the maturity of technology, the amount of interaction with data and the complexity of components attracted the least weight and were weakly correlated with the overall ratings. This suggests that these factors were not helpful in this particular exercise, though they may prove to be useful under different circumstances.

The second set of experiments focused on using an EA to find OWA weights that allowed us to aggregate overall ratings of the difficulty of attacking/evading security components to produce rankings of attacks for each expert that are strongly correlated with the actual attack rankings that they provided. This shows that we can extrapolate from ratings of individual components to rankings of complete attacks, lightening the load on experts.

It was also shown that if the participants are split arbitrarily, OWAs that work well with one group tend to well with the other too. This is a useful result, as it alludes to the possibility of using OWAs to aggregate unseen experts' hop ratings to produce ratings and rankings of unseen attacks for a proposed system. Comprehensive experiments showed that from 30 different starting positions (i.e., seeds) the EA consistently found similar solutions, suggesting that they represent globally good solutions, and that in all 30 runs the EA never placed less than 92% of the weight on the most difficult component. From this, we can infer that when assessing the difficulty of an attack, the most difficult to attack/evade security component is by far the most important. Simply taking the maximum proves impractical however, as it results in many attacks being given the same rating and ranking.

The final set of experiments combined the processes involved in the first and second sets of experiments. The best WAs and OWA found in previous experiments were used together to produce rankings of attacks from experts' ratings of factors that contribute to the difficulty of a component. The rankings produced, on average, show strong correlation with experts' actual rankings. This indicates that by asking targeted questions about characteristics of security components, specific technical attacks on proposed systems can be rated and ranked in terms of difficulty and salience. From a machine learning point of view, the proposed methodology where the parameters (i.e. weights) of aggregation operators (both WA and OWA) are optimised using an EA is clearly not the only viable approach. Different optimisation strategies such as simulated annealing could be explored, while the actual aggregation operators could be generalised for example to fuzzy integrals (Grabisch et al., 2000) where more complex weights captured by fuzzy measures enable a more fine-grained weighting.

These results raise the possibility of using ratings of security components to produce ratings and rankings for unseen attacks on proposed information systems. This could take the form of a database of ratings of generic security components (we have seen that lack of technical detail does not prevent reasoned assessment) which could be used for "What if?" scenarios to give an approximate assessment of the difficulty of attacks when security components are altered or moved. The degree to which this exciting possibility is feasible in practice is a matter of future research.

The major contribution of this paper to cyber-security research lies in its use of exploratory data analysis to demonstrate the feasibility of linking the experts rating of components of attacks to the overall ranking of the difficulty of the attack vectors themselves. In doing so, it both confirms the technical methodology of using WAs and OWAs to perform the aggregation, as well as confirming that the experts' ratings and rankings do have internal consistency with each other - which is not a *a priori* given, and in some ways can therefore be taken as refutation of our informally posed null hypothesis (that the security experts are not really experts). Hence, not only do these methods provide practical possibilities for systematising the analysis of attack vectors, but they also offer the future possibility of being used as some form of professional competence test (or accreditation) of security expertise. The results of this work also have the potential to reduce both the cost and time required to perform similar estimates through automation. However, we emphasise that these practical benefits will require further research and development. We are not aware of similar approaches to this problem being reported in literature, making direct comparisons impossible. We hope that this paper will initiate many similar studies in the future.

9. Conclusions

In this work we have presented a method of finding good aggregation operators for creating rankings of technical attacks on a proposed system and ratings of security components, which has the added advantage of also highlighting salient factors and security components. To do this, we have used a data set collected during a decision making exercise at GCHQ in which cyber-security experts performed security assessment tasks on a realistic system proposal. The results showed that using the proposed method, it is possible to produce rankings of technical attacks on a system using ratings of security components, ratings of security components from ratings a specific factors of difficulty, and finally, rankings of technical attacks on a system from ratings of specific factors. These outcomes present the possibility of using ratings of generic security components (or their characteristics) in "What if?" scenarios to assess the impact of altering/moving components within a system design.

The work has produced important insights that enable the construction of expert security assessment support tools. Such tools have the potential to reduce the time and effort required of experts for assessments, and enable systems designers to produce approximate security assessments before they seek expert advice. These advances will address growing concerns about the capacity of limited expert resources in view of the increasing complexity and growing number of information systems that are under attack from an ever changing set of attacks.

As noted in Section 8, this paper does not advocate the proposed approach of optimising the parameters of the given aggregation operators as a unique solution. Indeed, it is likely that other approaches will achieve similar, and in the case of more complex approaches such as fuzzy integrals, more nuanced results. However, the proposed architecture provides a viable approach which enables the generation of highly useful results and insight in the challenging area of expertled cyber security system assessment. The same general methodology may be applied to a variety of similar situations in which the modelling of expert opinion, particularly when aggregating sub-criteria, is desired.

Future work in this area will involve efforts to create the generic security component database mentioned previously and applying the techniques shown here to assess the practicality of providing decision support for systems architects in the process of designing new systems. In addition, we are exploring the applicability of the proposed approach to related areas such as "adversary characterisation" in cyber-threat intelligence.

Acknowledgements

The research reported here has been partially funded by CESG – the UK government's National Technical Authority for Information Assurance (IA) and RCUK's Horizon Digital Economy Research Hub Grant, EP/G065802/1. We also thank the anonymous reviewers who have made many insightful and valuable comments which have been incorporated into the text, thereby improving the paper.

Appendix A. EA configuration experiments

This appendix contains details of the experiments conducted to find ideal EA configurations for the OWA/WA search.

Appendix A.1. EA for use with OWAs

This subsection contains the experiments conducted to find an EA configuration to be used in the experiments performed in Section 6.

Initial experiments focused on discovery of an ideal configuration for the EA, which is a significant problem in itself. In order to create a practical test schedule that could be completed in a reasonable time frame, these experiments were split into two parts: 1) evolutionary operator proportions and 2) generations and population sizes.

Appendix A.1.1. Part 1 – evolutionary operator proportions

Table B.19 shows the configurations tested in the first part. For all of these experiments 1% elitism was applied to a population of 250 over 250 generations. The population and generations values have been chosen arbitrarily in order to explore the operator proportions. Appropriate values for population and generations will be examined in later experiments.

Table B.19 – EA-OWA: evolutionary operator proportion experiments.			
Test	Сору	Cross	Mut
1	0.00	0.20	0.79
2	0.20	0.20	0.59
3	0.40	0.20	0.39
4	0.60	0.20	0.19
5	0.79	0.20	0.00
6	0.50	0.00	0.49
7	0.30	0.40	0.29
8	0.20	0.60	0.19
9	0.10	0.80	0.09
10	0.00	0.99	0.00

Table B.20 – EA-OWA: evolutionary operator proportions – best results.			
Test	Mean Sp.	MSE	
Even			
3	0.6866	0.1425	
Odd			
1	0.5713	0.2543	
All			
1	0.6159	0.2069	
2	0.6159	0.2069	

The best results for these experiments are given in Table B.20. The results for all three groupings show that configurations that have a large proportion of mutation and a smaller proportion of crossover work best, and that the proportion of copy has little effect on the results. As the best individuals are automatically copied from one generation to the next via elitism, this may reduce the impact made by the copy operator. Table B.21 gives details of the OWAs found in each of the best experiments listed in Table B.20. It can be seen that in all cases, the EA is finding OWAs that place significantly more weight on the most difficult to attack/evade security component than the remaining components.

Appendix A.1.2. Part 2 – population and generations

In the second part of the EA configuration experiments, the focus switched to the number of generations and population size. The best evolutionary operator proportions from the first part of testing were used in a series of experiments with varying population sizes/numbers of generations. Table B.22 provides

Table B.21 – EA-OWA: evolutionary operator proportions – best OWAs.								
Test				Best w	veights			
	1	2	3	4	5	6	7	8
Even								
3 Odd	0.8985	0.0059	0.0311	0.0102	0.0065	0.0258	0.0037	0.0184
1 All	0.7872	0.0185	0.1242	0.0505	0.0111	0.0032	0.0029	0.0024
1 2	0.9582 0.9621	0.0028 0.0017	0.0242 0.0221	0.0003 0.0003	0.0057 0.0045	0.0029 0.0059	0.0002 0.0009	0.0056 0.0025

243

Table B.22 – EA-OWA: population and generations experiments.			
Test	Gens	Рор	
11	50	1250	
12	100	625	
13	200	315	
14	300	210	
15	400	155	
16	500	125	

the population sizes and numbers of generations tested. The best results from these experiments are shown in Table B.23. It should be noted that in the first part two configurations were joint best for the all group, both were tested in this part. The best result for the all group shown in Table B.23 was achieved with operator proportions from Test 2. In this set of experiments there is much less variation in the results than was seen in the first part of testing; altering the operator proportions had a greater effect on results than altering the population sizes and number of generations. Table B.24 gives details of the OWAs found in each of the best experiments listed in Table B.23. Again, the EA is finding OWAs that place significantly more weight on the most difficult to attack/evade security component than the remaining components.

Appendix B. EA for use with WAs

This subsection contains the experiments conducted to find an EA configuration to be used in the experiments performed in Section 5.

Table B.23 – EA-OWA: population and generations – best results.			
Test	Mean sp.	MSE	
Even			
12	0.6885	0.1421	
13	0.6866	0.1421	
Odd			
13	0.5732	0.2519	
16	0.5733	0.2519	
All			
15	0.6165	0.2080	

Table B.24 – EA-OWA: population and generations – best OWAs.								
Test	Best weights							
	1	2	3	4	5	6	7	8
Even								
12	0.8907	0.0011	0.0323	0.0120	0.0070	0.0174	0.0035	0.0359
13	0.8899	0.0044	0.0297	0.0126	0.0078	0.0242	0.0030	0.0284
Odd								
13	0.7858	0.0223	0.1285	0.0471	0.0115	0.0031	0.0015	0.0002
16	0.7854	0.0246	0.1248	0.0479	0.0117	0.0033	0.0001	0.0021
All								
15	0.9456	0.0022	0.0296	0.0006	0.0068	0.0011	0.0086	0.0055

Table B.25 – EA-WA: evolutionary operator proportion experiments.			
Test	Сору	Cross	Mut
1	0.00	0.20	0.79
2	0.20	0.20	0.59
3	0.40	0.20	0.39
4	0.60	0.20	0.19
5	0.79	0.20	0.00
6	0.50	0.00	0.49
7	0.30	0.40	0.29
8	0.20	0.60	0.19
9	0.10	0.80	0.09
10	0.00	0.99	0.00

Table B.26 – EA-WA: evolutionary operator proportions – best results.			
Test	Mean sp.	MSE	
Attack 2 Evade	0.7936	194.0279	
1	0.2270	420.2156	

Table B.27 – EA-WA: population and generations experiments.			
Test	Gens	Рор	
11	50	1250	
12	100	625	
13	200	315	
14	300	210	
15	400	155	
16	500	125	

Appendix B.1. Part 1 – evolutionary operator proportions

In the first set of experiments a series of evolutionary operator proportions are tested. As in the previous experiments, a fixed population of 250 is run over 250 generations and 1% elitism is applied. Table B.25 shows details of the configurations tested, the best results for each question set are shown in Table B.26.

These initial results show that the results are significantly different for each question set. The attack questions produce ratings that are extremely strongly correlated with experts' actual overall difficulty ratings, while the WAs for the evade questions produce ratings that are weakly correlated with experts' actual ratings.

Appendix B.2. Part 2 – population and generations

In the second part of the EA configuration experiments the best evolutionary operator proportions found for each question set in part 1 are used in a series of experiments with varying population sizes and numbers of generations. Table B.27 provides the configurations tested and Table B.28 shows the best results.

Again, the results for the attack question set are far superior to those for the evade questions.

Table B.28 – EA-WA: population and generations – best results.			
Test	Mean sp.	MSE	
Attack 15 Evade	0.7931	194.0125	
11–16	0.2270	420.2156	

REFERENCES

- Anderson R, Barton C, Böhme R, Clayton R, van Eeten MJ, Levi M, et al. Measuring the cost of cyber-crime. In: Böhme R, editor. The economics of information security and privacy. Berlin Heidelberg: Springer; 2013. p. 265–300 ISBN 978-3-642-39497-3, doi:10.1007/978-3-642-39498-0_12.
- Androutsopoulos I, Koutsias J, Chandrinos KV, Paliouras G, Spyropoulos CD. An evaluation of naive Bayesian anti-spam filtering. In: Potamias G, Moustakis V, van Someren M, editors. European conference on machine learning. Barcelona (Spain): 2000. p. 9–17.
- Badea A, Rocco C, Tarantola S, Bolado R. Composite indicators for security of energy supply using ordered weighted averaging. Reliab Eng Syst Saf 2011;96(6):651–62.
- Baskerville R. Information systems security design methods: implications for information systems development. ACM Comput Surv 1993;25(4):375–414.
- Bass T. Intrusion detection systems and multisensor data fusion. Commun ACM 2000;43(4):99–105.
- Canós L, Liern V. Soft computing-based aggregation methods for human resource management. Eur J Oper Res 2008;189(3):669– 81.
- Clements DP. Fuzzy ratings for computer security evaluation [Ph.D. thesis]. Berkeley, CA: University of California, Berkeley; 1977.
- Cruz B, Gupta D, Kapoor A, Haifei L, McLean D, Moreno F, et al., McAffee labs threats report, Tech. Rep., McAffee Labs, http://www.mcafee.com/us/resources/reports/rp-quarterly -threat-q1-2014.pdf>; 2014 [accessed 04.07.16].
- Detica and Office of Cyber Security and Information Assurance. The cost of cyber crime, Tech. Rep., Detica Limited, <http://www.cabinetoffice.gov.uk/resource-library/cost -of-cyber-crime>; 2011.
- Dhillon G, Backhouse J. Current directions in IS security research: towards socio-organizational perspectives. Inf Syst J 2001;11(2):127–53.
- Dondo M. A fuzzy risk calculations approach for a network vulnerability ranking system, Defence R&D Canada Technical Memorandum, 2007.
- Feyereisl J, Aickelin U. Privileged information for data clustering. Inf Sci (Ny) 2012;194:4–23.
- Goldberg D. Genetic algorithms in search, optimization, and machine learning. Reading Menlo Park: Addison-Wesley; 1989.
- Grabisch M, Murofushi T, Sugeno M. Fuzzy measures and integrals: theory and applications. Heidelberg: Physica-Verlag, 2000.
- Holland J. Adaptation in natural and artificial systems: an introductory analysis with applications to biology, control, and artificial intelligence. Ann Arbor: University of Michigan; 1975.
- Imamverdiev Y, Derakshande S. Fuzzy OWA model for information security risk management. Automat Control Comput Sci 2011;45(1):20–8.
- Jansen W, Gallagher PD. Directions in security metrics research, NIST Technical Report, 2009.

- Kim J, Bentley P, Aickelin U, Greensmith J, Tedesco G, Twycross J. Immune system approaches to intrusion detection – a review. Nat Comp 2007;4:413–66.
- Lane TD. Machine learning techniques for the domain of anomaly detection for computer security [Ph.D. thesis]. Lafayette, IN: Perdue University, 1998.
- Linkov I, Satterstrom F, Kiker G, Batchelor C, Bridges T, Ferguson E. From comparative risk assessment to multi-criteria decision analysis and adaptive management: recent developments and applications. Environ Int 2006;32(8):1072– 93.
- McGill WL, Ayyub BM. Multicriteria security system performance assessment using fuzzy logic. J Def Model Simul 2007;4(4):356–76.
- Merigó J, Gil-Lafuente A. New decision-making techniques and their application in the selection of financial products. Inf Sci (NY) 2010;180(11):2085–94.
- Merigó J, Gil-Lafuente A. Decision-making in sport management based on the OWA operator. Expert Syst Appl 2011;38(8):10408–13.
- Miller S, Garibaldi JM, Appleby S. Evolving OWA operators for cyber security decision making problems. 2013 IEEE Symp Comput Intell Cyber Secur 2013a;15–22. doi:10.1109/ CICYBS.2013.6597200.
- Miller S, Appleby S, Garibaldi JM, Aickelin U. Towards a more systematic approach to secure systems design and analysis. Int J Secur Softw Eng 2013b;4(1):11–30.
- Nettleton D, Torra V. A comparison of active set method and genetic algorithm approaches for learning weighting vectors in some aggregation operators. Int J Intell Syst 2001;16(9):1069–83.
- Ngai EWT, Wat FKT. Fuzzy decision support system for risk analysis in e-commerce development. Decis Support Syst 2005;40(2):235–55.
- Pearson K. Contributions to the mathematical theory of evolution. III. Regression, heredity, and panmixia. Proc R Soc Lond (Biol) 1895;59(353–358):69–71.
- Sadiq R, Rodrguez M, Tesfamariam S. Integrating indicators for performance assessment of small water utilities using ordered weighted averaging (OWA) operators. Expert Syst Appl 2010;37(7):4881–91.
- Shah S. Measuring operational risk using fuzzy logic modeling, International Risk Management Institute, Inc. (IRMI), <http://www.irmi.com/Expert/Articles/2003/Shah09.aspx>; 2003 [accessed 04.07.16].
- Siraj A, Bridges SM, Vaughn RB. Fuzzy cognitive maps for decision support in an intelligent intrusion detection system. Joint 9th IFSA World Congress 20th NAFIPS Int Conf 2001;4:2165–70.
- Spearman C. The proof and measurement of association between two things. Am J Psychol 1904;15(1):72–101.
- Sun L, Srivastava RP, Mock TJ. An information systems security risk assessment model under the Dempster–Shafer theory of belief functions. J Manage Inform Syst 2006;22(4):109–42.
- Tedesco G, Aickelin U. Real-time alert correlation with type graphs. In: Proceedings of the 4th International Conference on Information Systems Security (ICISS 2008). Berlin Heidelberg: Springer; 2008. p. 173–87.

- Torra V. The WOWA operator: a review. In: Yager RR, Kacprzyk J, Beliakov G, editors. Recent developments in the ordered weighted averaging operators: theory and practice, vol. 265. Berlin Heidelberg: Springer; 2011. p. 17–28.
- Tregear J. Risk assessment. Inform Secur Tech Rep 2001;6(3):19– 27. ISSN 1363-4127.
- Tukey JW. The future of data analysis. Ann. Math. Stat. 1962;33(1):1–67.
- Twycross J, Aickelin U. Information fusion in the immune system. Inform Fusion 2010;11(1):35–44.
- Van de Walle B, Rutkowski A-F. A fuzzy decision support system for IT service continuity threat assessment. Decis Support Syst 2006;42(3):1931–43.
- Yager R. On ordered weighted averaging aggregation operators in multicriteria decisionmaking. IEEE Trans Syst Man Cybern 1988;18(1):183–90.

Simon Miller currently works in industry as an analyst, specialising in modelling consumer behaviour. Prior to this he was a Research Fellow within the IMA research group, University of Nottingham, UK. The focus of his research is the development and application of Computational Intelligence techniques in realworld uncertain environments. He was awarded a PhD in Computer Science in 2011 from De Montfort University, Leicester, UK. Recently, he worked in collaboration with GCHQ on research that measured and modelled the variation that exists in the decision making of cyber-security experts. Dr Miller has published numerous articles in leading international journals and conferences.

Christian Wagner received the Ph.D. degree in Computer Science from the University of Essex, UK, in 2009. He is currently an Associate Professor of Computer Science at the University of Nottingham, UK. His research interests include the capture, modelling, interpretation and aggregation of uncertain data, the development of novel technique and their application in interdisciplinary contexts. Dr Wagner received Outstanding Paper Award for IEEE Transactions of Fuzzy Systems in 2013 and Best Paper Award at 2012 IEEE International Conference on Fuzzy Systems. He is a vice-chair of the IEEE CIS Task Force on Affective Computing and an Associate Editor of IEEE Transactions on Fuzzy Systems.

Uwe Aickelin is Professor of Computer Science and Vice-Provost for Teaching and Learning at The University of Nottingham Ningbo China. He received his PhD from the University of Wales (Swansea) in 1999. His research interests are Computational Modelling, Artificial Intelligence and Complexity Science for Data Analysis in the Health, Security and Digital Economy Domains.

Jonathan M. Garibaldi received the B.Sc. (Hons) degree in Physics from Bristol University, UK in 1984, and an M.Sc. degree (1990) and a Ph.D. degree (1997) from the University of Plymouth, UK. He is Professor of Computer Science, and leads the Intelligent Modelling and Analysis (IMA) Research Group, School of Computer Science, University of Nottingham, UK. His main research interests are modelling uncertainty and variation in human reasoning, and modelling and interpreting complex data to enable better decision making, publishing over 200 papers. He serves regularly in organising and programme committees of a range of leading international conferences.